\begin{table}%t1
\caption{Data used in this work.}
%\centerline
{\small
\begin{tabular}{lcccccc}
\hline\hline
Source &    $l$ &   $b$    & \EBV\tablefootmark{a} & \tauhi\tablefootmark{b} & \tauhcop\tablefootmark{c} & \WCO\tablefootmark{d} \\
       & $\circ$ & $\circ$ & mag      & \kms       & \kms &      \Kkms \\
\hline
   B1748-253& 3.745& 0.635& 7.86& 45.37(0.40)& &  \\
   B2005+403& 6.816& 4.302& 0.69& 4.67(0.05)& 0.41(0.02)& $<$0.20 \\
   B1730-130& 12.032& 10.812& 0.53& 10.91(0.17)& 1.16(0.02)& 0.47(0.12) \\
   B1908-210& 16.857& $-13.219$& 0.28& & $<$0.30& $<$0.20 \\
   B1819-131& 17.910& 0.372& 7.99& 63.71(1.00)& &  \\
   B1817-098& 20.711& 2.293& 1.55& 20.62(0.49)& &  \\
   B1819-096& 21.047& 1.957& 3.08& 29.01(0.40)& &  \\
   B1829-106& 21.347& $-0.629$& 11.56& 66.05(0.53)& &  \\
   B1741-038& 21.591& 13.128& 0.58& 8.20(0.11)& 1.11(0.10)& 1.11(0.07) \\
   B1849+005& 33.498& 0.194& 16.93& 117.62(0.87)& &  \\
   B1749+096& 34.920& 17.644& 0.09& & $<$0.14& $<$0.20 \\
   B1909+049& 39.694& $-2.244$& 2.51& 19.23(0.22)& &  \\
   B1910+052& 40.100& $-2.336$& 2.10& 17.68(0.41)& &  \\
   B1843+098& 41.112& 5.772& 0.57& 6.71(0.16)& &  \\
   B1915+062& 41.605& $-2.928$& 1.43& 12.78(0.27)& &  \\
   B1909+161& 49.658& 2.907& 1.54& 15.98(0.33)& & 4.47(0.06) \\
   B1905+190& 52.496& 5.591& 0.66& 9.69(0.23)& &  \\
   B1923+210& 55.557& 2.264& 1.87& 21.01(0.28)& 2.28(0.12)& 2.66(0.07) \\
   B1950+253& 62.366& $-0.956$& 3.15& 29.38(0.33)& &  \\
   B1901+319& 63.029& 11.757& 0.12& 0.42(0.10)& &  \\
   B1641+399& 63.455& 40.948& 0.04& & $<$0.09& $<$0.20 \\
   B2145+067& 63.656& $-34.072$& 0.08& 0.98(0.02)& 0.23(0.07)& $<$0.20 \\
   B2007+249& 64.048& $-4.334$& 1.27& 14.12(0.21)& & 16.62(0.25) \\
   J2023+319& 71.397& $-3.093$& 1.06& 12.11(0.08)& 1.55(0.04)&  \\
   B2015+33A& 72.226& $-0.978$& 3.48& 46.23(0.34)& &  \\
   B2015+33B& 72.226& $-0.981$& 3.48& 37.55(0.43)& & 5.31(0.11) \\
   B2023+336& 73.129& $-2.368$& 2.08& 22.90(0.26)& 3.43(0.02)& 6.04(0.04) \\
   B2048+313& 74.585& $-8.045$& 0.22& 1.43(0.22)& &  \\
   B2013+370& 74.866& 1.224& 1.78& 39.67(0.60)& 3.12(0.02)& 1.49(0.06) \\
   B1954+513& 85.298& 11.757& 0.15& & 1.68(0.06)& 1.89(0.04) \\
   B1823+568& 85.739& 26.080& 0.06& & $<$0.20& $<$0.20 \\
   B2251+158& 86.111& $-38.184$& 0.11& 1.78(0.01)& 0.30(0.01)& 0.91(0.04) \\
   B2037+511& 88.808& 6.040& 1.02& 17.32(0.13)& 0.65(0.17)& 0.51(0.06) \\
   B2022+542& 90.093& 9.665& 0.35& 4.17(0.16)& &  \\
   B2055+508& 90.378& 3.533& 2.95& 36.22(0.83)& & 19.65(0.12) \\
   B2106+494& 90.528& 1.305& 2.97& 54.15(0.52)& & 15.53(0.09) \\
   B2030+547& 91.129& 8.988& 0.38& 3.73(0.20)& &  \\
   B2200+420& 92.590& $-10.441$& 0.33& 3.74(0.04)& 2.36(0.03)& 6.78(0.05) \\
   B2154+483& 95.584& $-4.860$& 0.53& 7.51(0.13)& &  \\
   B2111+620& 100.287& 9.429& 0.84& 10.72(0.18)& &  \\
   B2146+608& 102.570& 5.713& 0.85& 11.04(0.11)& &  \\
   B2201+62S& 104.940& 5.833& 0.73& 9.44(0.14)& &  \\
   B1928+738& 105.625& 23.541& 0.13& & 0.73(0.03)& $<$0.20 \\
   B2341+535& 112.952& $-7.745$& 0.33& 5.62(0.09)& &  \\
   B2255+702& 113.596& 9.707& 0.52& 5.27(0.10)& &  \\
   B2357+554& 115.718& $-6.503$& 0.31& 4.65(0.23)& &  \\
   B2357+55B& 115.719& $-6.498$& 0.31& 2.27(1.10)& &  \\
   B2348+644& 116.513& 2.555& 1.26& 31.56(0.17)& &  \\
   B0012+610& 118.548& $-1.264$& 1.65& 34.20(0.15)& & 0.59(0.06) \\
   B0016+731& 120.644& 10.728& 0.32& 1.99(0.14)& &  \\
   B0041+660& 122.253& 3.449& 2.18& 44.63(0.32)& & 25.80(0.11) \\
   B0052+681& 123.351& 5.503& 1.00& 10.79(0.06)& &  \\
\hline
\end{tabular}}
\tablefoot{\tablefoottext{a}{Reddening from \cite{SchFin+98} rms error is 16\%;}
\tablefoottext{b}{\tauhi\ from \cite{GarDic89} and (for J-sources) this work see Sect.~2;}
\tablefoottext{c}{\tauhcop\ from \cite{LucLis96}, \cite{LisLuc00} and this work see Sect.~2;}
\tablefoottext{d}{\WCO\ from \cite{LucLis96}, \cite{LisLuc98} and this work see Sect.~2.}}
\end{table}